Author's response to reviews

Title: Pediatric integrative medicine: pediatrics' newest subspecialty?

Authors:

Sunita Vohra (svohra@ualberta.ca)
Soleil Surette (ssurette@med.ualberta.ca)
Deepika Mittra (dmittra@openingspaces.ca)
Lawrence D Rosen (ldrdoc@alum.mit.edu)
Paula Gardiner (Paula.gardiner@bmc.org)
Kathi J Kemper (kkemper@wfubmc.edu)

Version: 3 Date: 3 April 2012

Author's response to reviews:

Dear Ms. Neilan,

We would like to re-submit our revised manuscript (MS: 1047883178586099, Pediatric integrative medicine: pediatrics' newest subspecialty?) for consideration as a publication in BMC Pediatrics.

We, the authors, revised the methods section of our manuscript so that sufficient detail about how we conducted your study is provided. We added more information to the methods section and specified the study inclusion criteria.

We look forward to your reply.

Yours sincerely,

Sunita Vohra MD FRCPC MSc
Director, CARE Program for Integrative Health and Healing
Director, PedCAM Network
Professor
AHFMR Health Scholar
Dept of Pediatrics
Faculty of Medicine & Dentistry and School of Public Health
University of Alberta
8B19-11111 Jasper Ave
Edmonton General Hospital
Edmonton, AB
Canada
T5K 0L4
Phone: (780) 342-8592
Fax: (780) 342-8464